

Temi di Discussione

(Working Papers)

Academic performance and the Great Recession

by Effrosyni Adamopoulou and Giulia Martina Tanzi







Temi di discussione

(Working papers)

Academic performance and the Great Recession

by Effrosyni Adamopoulou and Giulia Martina Tanzi

Number 970 - September 2014

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Giuseppe Ferrero, Pietro Tommasino, Piergiorgio Alessandri, Margherita Bottero, Lorenzo Burlon, Giuseppe Cappelletti, Stefano Federico, Francesco Manaresi, Elisabetta Olivieri, Roberto Piazza, Martino Tasso. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Designed and printed by the Printing and Publishing Division of the Banca d'Italia

ACADEMIC PERFORMANCE AND THE GREAT RECESSION

by Effrosyni Adamopoulou* and Giulia Martina Tanzi†

Abstract

In this paper we study how the Great Recession affected university students in terms of performance, with a focus on the drop-out probability. To do so, we use individual-level data on a representative sample of university students in Italy in 2007 and 2011. We measure the severity of the recession in terms of increases in the adult and youth unemployment rates and we exploit geographical variation to achieve identification. On the one hand, an increase in the adult male unemployment rate worsens the financial condition of the family, raising the drop-out probability. On the other hand, by reducing the opportunity cost of tertiary education, an increase in the youth unemployment rate reduces the drop-out probability. Focusing on students who were enrolled at university before the Recession we are able to study the effects of the crisis on performance net of any potential effect on enrolment. We find evidence that overall, university drop-out decreased as a result of the Recession and that the probability of graduating on time increase for more motivated students.

JEL Classification: D12, E32, J24.

Keywords: academic performance, drop-out, Great Recession, unemployment.

1.	Introduction	5
2.	Data	9
3.	Empirical exercise	11
	3.1 Specification and main results	13
	3.2 Heterogeneous effects	15
	3.3 Gender differences	16
4.	Robustness	17
	4.1 Refining the measurement of the unemployment rate	17
	4.2 Endogeneity	17
	4.3 Common trend assumption	18
	4.4 Asymmetric effects of the business cycle	19
	4.5 University premium	19
	4.6 Other outcomes	21
5.	Conclusions	22
Та	bles	24
A	ppendix	40
Re	ferences	44

Contents

^{*} Bank of Italy, Directorate General for Economics, Statistics and Research.

[†] Bank of Italy, Economic Research Unit, Milan Branch.

1 Introduction¹

Labour market prospects and the financial situation of the family are known determinants of educational decisions (Becker, 1964). Job insecurity during the Great Recession (hereafter the Recession) might have changed the incentives to accumulate human capital. In the US there is evidence that the Recession increased university enrolment (Long, 2014). Little is known though on how university students reacted to this shock in terms of effort and the probability of dropping out. Although both the enrolment and the dropout decision are part of the decision to invest in human capital, dropping out entails one extra cost: the time and resources already invested in tertiary education. In this paper we use individual-level data on Italian university students who enrolled before the Recession and suffered its effects in their 2nd year of university studies, in order to examine the effect of the Recession on academic performance. We focus on three distinct measures of performance, i.e., the drop-out probability, the probability of graduating on time (i.e., 3 years after enrolment), and the frequency of course attendance.

The reason we focus on Italy is threefold. First, the availability of individual-level data that are nationally-representative and allow us to compare multiple student cohorts, enables us to exploit regional variation for identification, and provides us with information on the students' field and university of study, family background and ability measures, even for those who drop out. This kind of information is generally unavailable in population or labour force surveys. Second, the fact that Italy is among the European Union countries with the lowest percentage of university graduates and the highest youth unemployment rate. Third, the fact that contrary to the US, the Recession in its initial stage (2008-2010) was a shock that came to Italy from abroad.² Differently from Long

¹First version: February 2013. Many thanks to Antonio Accetturo, Jerôme Adda, Manuel Bagües, Lorenzo Burlon, Francesca Carta, Federico Cingano, Fabrizio Colonna, Francesco D'Amuri, Juan José Dolado, Daniel Garcia, Andrea Ichino, Francesco Manaresi, Vincenzo Mariani, Pedro Mira, Elisabetta Olivieri, Phil Oreopoulos, Evi Pappa, Chiara Pronzato, Jean-Marc Robin, Alfonso Rosolia, Jesse Rothstein, Diego Scalise, Paolo Sestito, Luigi Federico Signorini, Marco Tonello, and Roberta Zizza. We would also like to thank, for their useful discussions, the seminar participants at the European University Institute, the Bank of Italy, the Universidad Carlos III de Madrid, the University of Patras, the 5th IWAEE in Catanzaro, the 1st RWI Research Network Conference on the Economics of Education in Berlin, the 17th IEA World Congress at the Dead Sea, the 7th COSME-FEDEA Workshop on Gender Economics in Madrid, the IV Workshop on Public Policies, Social Dynamics and Population in Milan, and the 12th Conference on Research on Economic Theory, and Econometrics in Naxos. The views expressed in the paper are those of the authors and do not necessarily reflect those of the Bank of Italy. All the remaining errors are ours.

 $^{^{2}}$ See the discussion in Section 4.2.

(2014), we can consider the Recession as an exogenous "treatment".

The university drop-out rate has received a lot of attention in Italy because it has been very high. Since the 1970s and until the early 1990s it reached values well above 60 per cent (Cingano and Cipollone, 2007). In a related paper before the Recession, Di Pietro (2006) finds that regional unemployment rates are negatively associated with the drop-out rate. We focus our analysis on the latest cohort of students and we examine whether the Recession had a causal effect on the probability of dropping out of university.

Our identification strategy is based on regional variation in the severity of the Recession. We measure this in terms of the change in the adult male unemployment rate and the change in the youth unemployment rate between 2005-2007 and 2008-2010.³ The first is a proxy of changes in the financial situation of the student's family while the second is a proxy of changes in the opportunity cost of studying. We estimate a linear probability model with regional fixed effects using data on university students in Italy and we find that youth and adult unemployment rates have opposite effects on the probability of dropping out of university. The coefficient of the youth unemployment rate is negative and statistically significant, suggesting that the drop-out probability decreases as the opportunity cost of studying goes down. By contrast, the coefficient of the adult male unemployment rate is positive and statistically significant, and is larger in size. An adverse employment shock to the family of origin increases the drop-out probability due to financial constraints. Given that the unemployment rate has increased more sharply for the young during the Recession, the net effect is a decrease in the drop-out probability especially for men.⁴ A placebo test confirms that the estimated effects are not due to preexisting trends.

Our findings are in line with Aguiar, Hurst, and Karabarbounis (2013) who document that time spent in education by men in the US increased during the Recession. Using the cross-state variation in foregone market work they find that singles in general allocate more than 10 per cent of their increased time to education. Our results are also consistent with earlier papers on the countercyclicality of college enrolment in the US (Betts and McFarland, 1995; Dellas and Sakellaris, 2003). However, there is no direct correspondence

 $^{^{3}}$ The Recession hit Italy in the last quarter of 2008 and had adverse effects on labour markets (See D'Amuri, 2011).

 $^{^{4}}$ This gender difference depends on the students' field of study and on the educational level of the mother. See Section 3.3 for details.

between enrolment and drop-out as the latter entails the sacrifice of initial resources. Our data allow us to identify the effect of the Recession on the probability of dropping out net of any possible enrolment effect.

Recent papers shed light on the negative effects of graduating from college in a recession in the US (Kahn, 2010) and Canada (Oreopoulos, von Wachter and Heisz, 2012). Both papers find negative wage effects that persist over time. Hershbein (2012) focuses on high school graduates in the US and finds that their wages were less affected by the Recession than those of college graduates. We focus instead on the educational outcomes of university students and we find that the Recession may also affect human capital accumulation through changes in the university drop-out probability.

The idea that the Recession affects academic performance can be viewed as part of the broad literature on credit constraints in education. Lovenheim (2011) and Lovenheim and Reynolds (2012) find that households used their housing wealth to finance college enrolment in the 2000s when housing wealth was most liquid. Their findings imply that the recent housing bust could significantly affect college enrolment and completion through the reduction in the housing wealth of families with college-age children. Bound, Lovenheim, and Turner (2010) find that a rise in the housing wealth of the student's family increases the probability that the student will attend a better university (flagship public university instead of a non-flagship one). They also find some evidence of an increase in the likelihood of completing college for lower income students. Cameron and Taber (2004) instead do not find evidence that borrowing constraints generate inefficiencies in the market for schooling. Their identification is based on the prediction that the opportunity cost of schooling (measured by local low-skill wage rates) and the direct cost (measured by whether there is a college in the individual's county of residence) affect borrowing-constrained and unconstrained persons differently.

Credit constraints may be confused with adverse initial conditions. Carneiro and Heckman (2002) find that long-run factors reflected in ability are the major determinants of the family income-schooling relationship, and that only 4 per cent of the US population is credit-constrained in the short-run. Belley and Lochner (2007) find that family income has become a much more important determinant of college attendance in the early 2000s than in the 1980s suggesting that credit constraints might be more relevant for the recent cohorts. The richness of our data allows us to control for initial conditions by providing us with information on the parents' education, labour market status, and occupation when the students were 14 years old.

Credit constraints can be a reason for dropping out of university. Stinebrickner and Stinebrickner (2008) use direct information on the reasons given by students for dropping out of a college in Kentucky, and examine how many students would drop out even if credit constraints were removed. They find that although credit constraints are likely to play an important role in the drop-out decisions of some students, the large majority of students from low income families drop out primarily due to reasons other than credit constraints. In a more recent paper using the same data, Stinebrickner and Stinebrickner (2012) find that college drop-out occurs as students learn about their academic ability or grade performance after matriculation. Our database, which is representative of the entire population of university students in Italy, also contains information on the reasons for dropping out, with "studies being too costly" among the possible reasons. In our data, both before and during the Recession, only around 5 per cent of university students dropping out report "studies being too costly" as the main reason.

Students' college preparation as well as collegiate characteristics may also affect the drop-out decision (Bound, Lovenheim, and Turner, 2010). We use high school grade as a measure of students' ability. We also exploit official data on the ranking of Italian universities by field of study. We can thus control for the ability of the students and the quality of the university they attend.

There are also papers studying the relationship between local labour market conditions and the probability of leaving post-compulsory secondary education before the Recession (see Petrongolo and San Segundo, 2002 for Spain; Clark, 2011 for the UK; Mocetti, 2012 for Italy). Given the young age of high school students, the decision to drop out might mainly reflect the will of the family rather than that of the student. By focusing instead on university students we are more likely to capture the decision to drop out being made by the students themselves. Moreover, secondary education in Italy is practically free of charge as most students attend public high schools, where there are no tuition fees. By contrast, tertiary education is costly, as both private and public universities charge tuition fees. The rest of the paper is organized as follows. Section 2 describes the data and presents evidence on regional differences in the severity of the Recession. Section 3 introduces the empirical strategy, discusses the identification issues, and presents the main findings. Section 4 includes robustness checks and the results for other outcome variables. Section 5 concludes.

2 Data

Our data come from the Survey on Educational and Professional Paths of Upper Secondary School Graduates conducted by the Italian National Institute of Statistics (Istat).⁵ The survey covers a representative sample of high school graduates in Italy three years after high school graduation. There are 4 waves available: Survey 2001 on graduates of 1998, Survey 2004 on graduates of 2001, Survey 2007 on graduates of 2004, and Survey 2011 on graduates of 2007. The survey consists of more than 20,000 respondents in each wave and provides detailed information on educational and employment history as well as on parental background. High school graduates at the time of the survey might study at a university, work, study and work at the same time, be unemployed, or be inactive. In the case of drop-outs there are questions regarding educational history up to the moment of dropping out as well as information on the reason for dropping out.⁶

The last wave (Survey 2011) is the survey conducted during the Recession. Our aim is to compare students' academic achievement before and after the Recession. However, for administrative reasons, the last wave took place 4 years after high school graduation. This makes the comparability of the last wave with previous waves less straightforward. Exploiting the information on the exact time of dropping out, we focus on individuals who dropped out no later than their 3rd year of university studies. This ensures comparability with previous waves with respect to our key variables.

We construct our measures of the severity of the Recession using data from the Italian

 $^{^{5}}$ The analyses were conducted at the Istat-Italian Research Data Center (Laboratorio Adele) in compliance with confidentiality policies and procedures. The opinions expressed in this paper are the sole responsibility of the authors and do not represent the official position of the Italian National Institute of Statistics.

⁶This survey contains information that is not generally available in population or labour force surveys. First, the grade obtained at high school in order to control for student's ability. Second, the field and the university of study, including for those who have dropped out.

Labour Force Surveys in the period 2005-2007 and 2008-2010 (3-year averages corresponding to the academic years of each cohort). We compute the change in the unemployment rate for adult males aged 35-74 years old by educational attainment, and for young high school graduates aged 20-24 by gender in the student's region of origin (Figures A1 and A2). The first measure is a proxy of the current financial situation of the father.^{7,8,9} Contrary to the US, student loans in Italy are not a common practice (Brown and Sessions, 1999). In most cases parents finance the university studies of their children. In our data less than 15 per cent of university students combine work and study. Therefore, the employment situation of the father matters. The second measure is a proxy of the opportunity cost of studying. As Table 1 shows, in Italy student mobility is low between regions (around 20 per cent) but much higher between provinces (around 50 per cent). Given that the majority of students study in the region of origin we assign them with the corresponding unemployment rate. For those who study in a different region we assume that if they drop out before finishing university, they will look for a job in the region of origin as high school graduates. In the 2011 survey, the only wave which gives the region of a person's current job, a mere 8 per cent of university drop-outs found a job in a region different from the region of origin.^{10,11}

In order to control for the quality of the universities we merge our database with official data on quality indicators for all the universities in Italy by field of study. The data come from a research evaluation survey conducted in 2006 by the Supervising Committee for Research Evaluation (CIVR) with the collaboration of the Ministry of Education, Universities, and Research (MIUR). The measure of quality is an aggregate indicator that

⁷The survey does not contain information on the current situation of the parents. The only available information on parents' education, employment and occupational status refers to 8 years before the survey, i.e. to the time when the respondents were 14 years old. In order to proxy the current employment status of the father we assign to each student the adult male unemployment rate that corresponds to the educational group of his/her father in the region of origin.

⁸We focus on adult males since in Italy more than 50 per cent of adult females do not work.

⁹Our proxy does not need to capture parental layoffs. Our results go through as long as the perception of the father about the probability of layoff is affected. Narrowing the age group to 45-64 produces similar results.

¹⁰The figure is around 20 per cent for drop-outs who used to study in a region different from the region of origin.

¹¹In the analysis that follows our results hold, even if we exclude the students who study in a different region from their region of origin. Moreover, assigning each student with the youth unemployment rate in the region of study instead of the one in the region of origin produces similar results (available upon request).

takes into account the number and the rating of courses on offer, the average characteristics and the number of courses of excellence, the number of years spent by researchers in international mobility programs, the number of PhD and post-doctoral researchers, as well as the amount of research funds received from various sources (Ministry of Education, European Union, etc.).

Table 1 shows the descriptive statistics for the variables of interest for 3 different cohorts. We first focus on the last 2 columns (the cohort before the Recession and the cohort of the Recession). We observe that the drop-out rate slightly decreased between 2007 and 2011.¹² At the same time, the percentage of students who graduate on time and of those who attend classes more than 3 times a week increased. Both the adult male and the youth unemployment rate have risen during the Recession with the rise being larger for the young. Figures A1 and A2 show that there has been variation in the severity of the Recession across regions. We now move to the empirical exercise in order to examine whether the Recession has had any effect on students' performance.

3 Empirical exercise

According to the predictions of Becker's basic model on human capital accumulation (1964), the Recession has two opposite effects on academic performance. On the one hand, parental resources decrease as the unemployment rate of the fathers goes up. We expect this to positively affect students' performance in order to speed up graduation. On the other hand, the opportunity cost of studying decreases as the youth unemployment rate goes up. We expect this to negatively affect performance by delaying graduation. Regarding drop-out, we expect the opposite effects. The rise in the unemployment rate of the fathers is expected to increase drop-out due to financial difficulties, while the rise in the youth unemployment rate is expected to decrease drop-out due to the scarcity of outside options. In our data 76 and 67 per cent of university drop-outs in 2007 and 2011 respectively are working, suggesting that finding a job became more difficult during the crisis.

The Recession might also have influenced enrolment, with indirect effects on students'

¹²A recently released report from the National Agency for the Evaluation of Universities and Research Institutes (ANVUR, 2014) confirms the decrease in the drop-out rate.

performance. Less competent students who before the Recession would have preferred to work instead of going to university, might decide to go to university during the Recession due to a lack of job opportunities. Likewise, more competent students who before the Recession would have preferred to go to university might not be able to do so during the Recession due to lower parental income. As a result there would have been a composition effect which might have undermined performance. The timing of the survey enables us to study the effect of the Recession on performance net of any potential effect on enrolment. The most recent wave of the survey took place in 2011, i.e. during the Recession and involved young individuals who graduated from high school in 2007. The vast majority of those who enrolled at a university did so immediately after graduation in 2007, i.e. before the Recession hit Italy.¹³ Hence, these students' enrolment decision was not affected by the Recession. Given that the Recession had not been predicted (Bezemer, 2009), we can also rule out anticipation effects.

The university system in Italy underwent a major reform in 2001. In particular, a 3-year first-level degree followed by a 2-year second-level degree (3+2) replaced the old degree, typically of 4-year duration in most fields (Bratti, Broccolini and Staffolani, 2006). In our exercise we use only students in 3+2 programmes who enrolled at university after the reform (the earliest cohort of students in our sample started university in 2001). Moreover, tuition fees increased during the Recession. According to the available data for a group of universities (Federconsumatori Surveys 2010, 2011, 2012), university tuition fees increased around 10 per cent between 2011 and 2012 for students with a family income of less than 10,000 euros but remained practically unchanged in the period 2010-2011. In our sample we define as the "Recession cohort" the cohort of students who enrolled at university in 2007 and we analyze the probability of dropping out up to the year 2010. Hence, tuition-fees increases are unlikely to affect our results.

¹³In our analysis we only take into consideration young individuals who enrolled at university immediately after high school graduation.

3.1 Specification and main results

We start with a linear probability model with regional and cohort fixed effects.¹⁴ The surveys in 2007 and 2011 refer to two different cohorts, therefore we do not observe the same individual over time. We proxy the financial situation of the student's family with the adult unemployment rate for males aged 35-74 in the student's region of origin according to the education of his/her father. We also proxy the opportunity cost of the student with the youth unemployment rate for high school graduates aged 20-24 years old by gender in the student's region of origin. We start our analysis with the drop-out probability as the outcome variable.

Our benchmark specification is the linear probability model specified in (1),

$$dropout_{i,r,c} = \beta_o + \beta_1 (Adult \ male \ unemployment \ rate)_{r,c} + \beta_2 (Youth \ unemployment \ rate)_{r,c} + \beta_3 X_{i,r,c} + \beta_4 (cohort)_c + \beta_5 (region)_r + \epsilon_{i,r,c}$$
(1)

where *i* stands for the individual, *r* for the region, and *c* for the cohort. The dependent variable is discrete and takes the value 1 if the student dropped out of university and 0 otherwise. The independent variables are the adult male unemployment rate, the youth unemployment rate, time (cohort) dummies, regional dummies, and $X_{i,r,c}$ that includes individual controls, namely the gender, the school grade as a proxy of ability, a dummy for having a father with a high school diploma, a dummy for having a father with a university degree, an indicator for coming from a disadvantaged family in order to account for initial conditions,¹⁵ a dummy for studying in a private university and a dummy for studying in a different region from the region of origin. We also include dummies for the particular university of study and for the specific field of study (e.g. engineering, political sciences, etc.). Hence, we are able to control for the level of difficulty that differs across fields and

¹⁴This is equivalent to a Difference-in-Differences approach with continuous treatment where we compare the outcomes before and after the Recession and across treated and untreated individuals. This approach has the advantage that individuals are not strictly assigned to treated and untreated groups with a switch-on/switch-off dummy variable. Instead, the unemployment rate is used as a variable with differing treatment intensity across regions and cohorts. See Angrist and Pischke (2009) for more details.

¹⁵We define as students from disadvantaged families those whose father was either dead, an unqualified worker (active or retired), or unemployed, and whose mother was either dead, an unqualified worker (active or retired), unemployed, or a housewife.

universities, as well as for university-specific traits (tuition fees, teaching practices, etc.).

We first estimate the model without the individual controls $X_{i,r,c}$ (Table 2, Column 1). We find a positive and statistically significant effect of the adult male unemployment rate on the probability of dropping out. Moreover, as expected, there is another channel at work, that generated by the increase in the youth unemployment rate. Indeed, we find that the youth unemployment rate has a negative and statistically significant effect on the drop-out probability. These effects are robust to the inclusion of individual controls. Both coefficients decrease in magnitude but remain statistically significant (Table 2, Column 2). Going through the coefficients of the individual controls (Table A1), we find that the probability of dropping out is lower for women, for more able students (proxied by the high school grade), and for those whose father has a high level of education. Moreover, the drop-out probability is higher for students who come from a disadvantaged family, and it is lower for students that study away from home (in another region) or at a private university, probably because of the high initial sunk cost.¹⁶

In Columns 3 and 4 of Table 2 we estimate the same regressions (with and without individual controls) substituting the regional fixed effects with fixed effects that capture, at the same time, the region, the level of education of the father and the gender of the student. As previously explained, our proxy of the intensity of the Recession for the parents is the unemployment rate of the adults located in the same region, and with the same level of education as the student's father. Moreover, the youth unemployment rate varies not only according to the region of the student but also according to his/her gender. Including more specific dummies than the regional ones allows us to better disentangle the effects on drop-outs that can be attributed to the crisis. Results are completely in line with those found using simple regional dummies and in the complete specification (Table 2, Column 4) the coefficients of our variables of interest are even larger than those in Column 2.¹⁷

In order to see which effect dominates and to compute the net effect of the Recession on the drop-out probability, we need to consider the changes in the adult male and the youth

¹⁶We also repeat the analysis by including the unemployment rates one by one and the results are very similar (Table A2).

¹⁷In Column 2 the specification includes among the individual controls the gender of the student and the level of education of the father, so as to take into account these covariates even when we include simple regional dummies.

unemployment rates. As shown in Table 1 the adult male and the youth unemployment rates have increased in Italy by 0.99 and 2.36 percentage points between the period 2005-2007 and 2008-2010. Multiplying these numbers with the most conservative estimates from the linear probability model (Table 2, Column 2) we get a decrease in the dropout rate of 0.08 percentage points. This suggests that the opportunity cost channel outweighed that of the financial situation of the parents. Given that the drop-out rate between the two cohorts fell by 0.65 percentage points, the Recession explains 12.3 per cent of this drop. Considering that in the years that followed (2011-2013) the youth unemployment rate has further increased at a fast pace, the overall fall in the drop-out rate due to the Recession is potentially large for this specific cohort of students.

3.2 Heterogeneous effects

The changes in the opportunity cost of studying and in the financial situation of the family during the Recession may have affected students' drop-out probability in a heterogeneous way. We thus perform the same analysis by different groups according to individual characteristics (such as the student's gender, ability, and family background), and university characteristics (field of study and quality of the university).

As Table 3 shows, the effects of the Recession differ vastly according to gender. First, the increase in the adult male unemployment rate led to an increase in the drop-out probability only for women, while the effect is not statistically different from zero for men. Second, the increase in the youth unemployment rate led to a greater decrease in the drop-out probability for males than for females. These two findings imply that the net effect of the Recession was a decrease in the drop-out probability for males and an increase in the drop-out probability for females. In the next subsection we discuss possible explanations behind these gender differences.

We find that the Recession mainly affected less able students in terms of drop-out probability (Table 4). This is not surprising given that less able students are the students at most risk of dropping out. More able students instead remained unaffected.

There are also large differences according to paternal socioeconomic status (Table 5).¹⁸ For students coming from disadvantaged families we find a large and statistically

¹⁸Here, instead of splitting the sample we introduce interactions as the size of the disadvantaged

significant increase in the drop-out probability as the adult unemployment rate goes up. Moreover, for these students the change in the opportunity cost of studying does not seem to matter. As expected, families with a more vulnerable economic situation had more difficulties in sustaining the burden of the cost of education. This is consistent with Christian (2007) who finds more procyclical enrolment among people from lower-income, lower-education households in the US.

We then consider whether there are heterogeneous effects according to university characteristics. Table 6 presents the estimates of the probability of dropping out by field of study. The coefficient of adult male unemployment rate is positive for students in the humanities (political/social sciences, law, literature, languages) but negative for students in the sciences (math/physics, geology/biology, engineering/architecture, economics/statistics). Studying science might be considered by parents as an important investment for the future of their children and therefore drop-out decreases in spite of the worsening of financial conditions.

Lastly, we analyze the choices of students according to the quality of the university by field of study chosen. We find that the net effect of the crisis is a large increase in the drop-out probability for students in universities placed below the 25th percentile in the rankings, an increase smaller in magnitude for students in medium quality universities, and no significant increase for students in high quality universities (Table 7).

3.3 Gender differences

The analysis so far has revealed a disparity in the effects of the Recession on the drop-out decisions of young men and women. In this subsection we try to shed light on these gender differences. Why do women drop out more as adult male unemployment rises while men remain unaffected? One possible explanation is the field of study. Women are traditionally concentrated in the humanities while men are concentrated in the sciences. In Table 8a we analyze the behaviour of women and men by field of study. We find that the effect of adult male unemployment rate is positive for both genders in humanities. However, the effect is statistically significant only for women and much larger in magnitude. It seems that parents consider studying humanities more as a "luxury" for

students' group is small.

their daughters.

Another possible explanation lies in the education of the mother. Mothers may act as a role model for their daughters and as a result the choice to finish university or drop out might differ according to the educational attainment of the mother (See Cardoso, Fontainha, and Monfardini, 2010 for the pronounced role of the mother on children's human capital investment in Italy). In Table 8b we investigate this possibility. Indeed the positive effect of the adult male unemployment rate on the drop-out probability is counterbalanced for women whose mother has finished university. This means that an adverse financial shock to the father makes women drop out more especially in the case that their mothers are not university graduates.

4 Robustness

4.1 Refining the measurement of the unemployment rate

In our baseline specification we exploit regional variation in a continuous way in order to achieve identification. In Italy the prospective labour market for the young lies often beyond the borders of their province, and therefore it is more reasonable to proxy the employment conditions that the young face with those in the corresponding region. Nevertheless, we moved to provincial variation in order to proxy the opportunity cost and the financial situation of the family in a more detailed way.¹⁹ We obtained similar results (Table 9) but by disaggregating the unemployment statistics, we increased the measurement error. The labour force measures of unemployment for particular age groups, gender, and educational attainment are not representative at the provincial level and as a result our estimates are more noisy.

4.2 Endogeneity

One might worry that a Recession cannot be considered as an exogenous treatment. Indeed, if the origin of the Recession had been idiosyncratic and related to preexisting structural differences across regions, our results would not have been causal. The Reces-

¹⁹We also adjusted the adult unemployment rate to include workers on redundancy payments (Cassa Integrazione) but this did not change our results.

sion instead started with the US subprime mortgage crisis in 2007 and was transmitted to Europe in October 2008 (Bordo, 2008). Hence, it was a shock that came to Italy from outside its borders. Caivano, Rodano, and Siviero (2010) find that 75 per cent of the crisis that Italy experienced in the period 2008-2010 was imported from abroad while only 16 per cent can be attributed to internal financial factors or a lack of confidence. It is only after 2010 that structural problems started to play an important role.

Yet, how severely the different regions of Italy experience the Recession might depend on the economic and social conditions of each region. For example, the South of Italy was typically characterized by a high unemployment rate while the opposite was true for the North. However, as shown in Figures A1 and A2, contrary to what one would expect, the Recession had different effects on regions that shared similar characteristics in terms of economic development.

Another possible concern is that as university drop-outs enter the labour force the youth unemployment rate will become endogenous. In our analysis we focus on the 20-24 age group (instead of 19-21) in order to alleviate this concern. Most of the drop-out decisions take place in the 1st year of studies when the students are 19 years old. Moreover, dropouts represent a very small fraction of all young people in the 20-24 age group.

4.3 Common trend assumption

The Difference-in-Differences approach is based on the "common trend" assumption, i.e. that the underlying trends in the outcome variable are the same in treated and control groups. It is in general difficult to test whether this assumption is violated. In our case it is possible, since we have data on the 2004 cohort. We thus run a placebo regression using data on two cohorts before the Recession (the 2004 and 2007 surveys) but using the unemployment rates during the Recession (2005-2007 and 2008-2010). We estimate the same model as in (1) for the placebo exercise (Table 10) and we get statistically insignificant estimates, that are very small in magnitude and have the opposite coefficients to those in the benchmark true regression.²⁰ This reassures us that the estimated effects

 $^{^{20}}$ In the placebo regression we do not include university and field of study dummies as this information is not available for drop-outs in the 2004 survey. For the same reason we are not able to control for studying in a different region from the region of origin and for studying in a private university. We

of the unemployment rate on the drop-out probability are indeed due to the Recession and not due to preexisting trends.

4.4 Asymmetric effects of the business cycle

Our results so far indicate that the net effect of the Recession on the drop-out probability was to reduce it. One might wonder whether during booms the opposite is true. Going back to Table 1 we observe that between the periods 2002-2004 and 2005-2007 the adult male and youth unemployment rates had decreased. At the same time the drop-out rate had increased. We now estimate the same linear probability model for the probability of dropping out as in (1) using the 2004 and 2007 surveys and the corresponding unemployment rates (Table 11). For the earliest cohort the university and the field of study is not available for drop-outs so we are not able to include the corresponding dummies and the controls for private university and for studying in a different region in the estimated specifications. As soon as we control for individual characteristics we do not find any statistically significant effect of the boom on the dropout probability. It seems that the effect of the business cycle on academic performance is not symmetric. One possible explanation is that the Recession, in contrast to the previous business-cycle fluctuations, was an unexpected shock that students did not internalize when deciding to enroll at university.

We obtain similar results when we pool all 3 cohorts (the 2004, 2007, and 2011 surveys) and we interact the adult male and youth unemployment rates with a dummy for the Recession (Table 12). The coefficient of the adult male unemployment rate is statistically significant only when interacted with the Recession dummy. The youth unemployment rate seems to matter both in good and bad times but its effect increases further during the Recession.

4.5 University premium

In our analysis we proxied the opportunity cost of tertiary education using the youth unemployment rate of secondary-education graduates. However, a university student might also consider factors related to the university premium when deciding whether to exclude these variables from the benchmark true regression to ensure comparability. drop out or not. In particular, the probability of finding a job as a university graduate might differ from the probability of finding a job as a high school graduate. Likewise, the wage that one will earn as a university graduate might be different from the wage of a high school graduate.

In order to account for the university employability premium we compute the unemployment rate of university graduates aged 25-29 by region and gender using the Labour Force Survey. We define the university employability premium as the difference between the unemployment rate of university and high school graduates. Indeed, the employability of university graduates with respect to high school graduates has improved during the Recession (Table A3, upper panel).

Data on wages by education, gender, and age group at regional level are more difficult to find. The Labour Force Survey started collecting information on wages only in 2009. We use the Survey on University Graduates' Vocational Integration in 2007 and 2011 in order to compute the monthly net wage of university graduates three and four years after graduation by gender and region. Similarly, we draw information on the monthly net wage of high school graduates from the Survey on Educational and Professional Paths of Upper Secondary School Graduates in 2007 and 2011.²¹ We compute the university wage premium as the difference in the monthly net wage of university and high school graduates. It seems that the university wage premium has increased during the Recession (Table A3, lower panel).

We now re-estimate the model of the probability of dropping out including the university employability premium (Table 13, Column 1) or the university wage premium (Table 13, Column 2) as an extra regressor. None of the university premium variables is statistically different from zero. By contrast, the youth unemployment rate remains statistically significant. This result implies that during the Recession university students did not seem to consider the university premium in their decision to drop out. In fact, the increase in the youth unemployment rate has received a lot of attention by the Italian media. This might explain its key role in shaping the drop-out decisions.

 $^{^{21}\}mathrm{We}$ restrict the sample to high school graduates who never enrolled at university so as not include drop-outs.

4.6 Other outcomes

Apart from the effect on the drop-out rate, the Recession may also have altered the behaviour of the students that did not interrupt their studies. In particular, there might have been an effect on the timing of graduation and on the regularity of class attendance, which are both measures of the effort exerted by the student. The youth and adult male unemployment rates may have opposing effects on these outcomes as well. We expect a positive effect of an increase in the adult male unemployment rate. Parents may put pressure on their child to graduate on time so as to save money in terms of university fees or to be a more attractive candidate when entering the labour market. At the same time, an increasing youth unemployment rate may make students delay graduation, as job opportunities become scarce. Similarly for the regularity of class attendance. Indeed, we find a positive effect of adult unemployment rate on the probability of graduating on time and a negative effect of youth unemployment rate (Table 14).²² These effects though are not statistically significant when we consider all students (Table 14, Column 1). Moving to the analysis by different groups we consider those students that chose to study at a particular university because it was the most convenient in terms of distance (less motivated students) versus those that made the choice of university based on its prestige, and the quality of the services it offers (more motivated students).²³ The adult unemployment rate has a large positive effect on more motivated students. These students try to graduate as soon as possible as their families face financial difficulties. When we perform the analysis by university quality, we find a negative effect of the youth unemployment rate for students at low quality universities. Lastly, we do not find any effect of the Recession on the regularity of class attendance (Table 15).

 $^{^{22}}$ In our analysis we only consider students who are enrolled in 3+2 programmes, i.e. 3-year bachelor degree followed by a 2-year master. In 2011 the survey took place 4 years after university enrollment and we defined as on-time graduates those who graduated by the end of 2010. In 2007 the survey took place 3 years after enrollment but as the interviews were conducted at the end of 2007 (between November 2007 and February 2008), we have information on on-time-graduation.

 $^{^{23}}$ See Sestito and Tonello (2011) and Rizzica (2013) on the issue of student mobility and university choice in Italy.

5 Conclusions

This paper adds to the understanding of university drop-out by examining the effect of the Recession in Italy. The effect of a recession on the student drop-out rate is theoretically ambiguous. The decrease in the opportunity cost of studying due to the increased difficulty in finding a job may lead to a fall in the drop-out probability. At the same time, the worsening of the labour market conditions for adults may translate into more adverse financial conditions for the families. This may result in an increase in the drop-out probability due to the lower availability of funds. In order to capture the causal effect of the Recession, it is fundamental to disentangle its effect on students' drop-out decisions from its effect on enrolment. In order to do this, we used unique information on a cohort of university students in Italy who enrolled before the Recession and were affected by it during their second year of studies. This made this cohort perfectly comparable to a previous cohort of students who enrolled and completed (or dropped out from their studies before the occurrence of the Recession. We then explored regional variation in changes in the adult male and the youth unemployment rate, that proxied respectively the adverse financial conditions of the student's family and the falling opportunity cost of studying. We found evidence that increases in the adult unemployment rate have a positive effect on the drop-out probability of university students, while increases in the youth unemployment rate have a negative effect, albeit smaller in magnitude. Since the unemployment rate of the young has increased to a greater extent than the unemployment rate of adults, the net effect of the Recession is a fall in the drop-out probability that can be translated into an increase in human capital accumulation. Considering that in the years 2011-2013 the youth unemployment rate increased further, the net effect of the Recession is potentially larger for this specific cohort of students.

Understanding to what extent and through which channels a crisis may affect students' choices has important policy implications. Our results suggest that human capital accumulation increased during the first years of the Recession. This is of particular importance as human capital can lead to economic growth (Ciccone and Papaioannou, 2009), a key feature in order to exit the Recession. Nevertheless, before drawing conclusions, one should consider any other possible side effects like over-education or the possibility that students are merely "parking" themselves at the university. Despite our finding that on-time graduation increased for more motivated students, our data did not allow us to observe whether the students who did not drop out managed to graduate eventually. Even in the case that the number of graduates increases, it is not clear whether the Italian labour market will be able to absorb them and if so in positions that match their qualifications. Moreover, the effect of the Recession was not homogeneous across different socioeconomic groups. The drop-out rate increased sharply for students from disadvantaged families as a result of the Recession. In a period of economic turbulence policy makers should make scholarships available to students in this specific target group.

Tables

Survey year year of enrolment at university	$\underset{2001}{2004}$	$\underset{2004}{2007}$	$\underset{2007}{2011}$
% drop-out	11.41	13.72	13.07
% on-time graduates	n.a.	14.59	18.50
% students attending class >3 times/week	90.96	89.53	93.73
% students working while studying	10.19	14.91	13.29
adult male unemployment rate (3-year average)	2.83	2.61	3.60
	(2.75)	(1.94)	(2.43)
youth unemployment rate (3-year average)	24.16	20.05	22.41
	(18.14)	(12.09)	(11.17)
% female	55.11	56.65	57.26
average high school grade	80.38	82.46	81.07
	(12.55)	(13.00)	(12.64)
% from disadvantaged families	18.24	19.58	18.96
% with father university graduate	18.02	16.61	18.41
% with father high school graduate	42.64	45.05	46.08
% who study at a private university	n.a.	5.66	6.36
% study in a different region from that of origin	n.a.	17.20	18.22
% study in a different province from that of origin	n.a.	48.15	47.22
Ν	8,060	$12,\!835$	12,419

Table 1. Descriptive statistics of the working sample, Mean (standard error)

Notes: corrected for the survey design using the corresponding weights.

The working sample includes university students or drop-outs in 3+2 programmes, who enrolled at university immediately

after high school graduation

	(1)	(2)	(3)	(4)
Adult male unemployment rate	0.020***	0.004*	0.010*	0.008*
	(0.002)	(0.002)	(0.005)	(0.004)
Youth unemployment rate	-0.009***	-0.002**	-0.005***	-0.003***
	(0.001)	(0.001)	(0.001)	(0.001)
Individual controls	No	Yes	No	Yes
Cohort dummies	Yes	Yes	Yes	Yes
Field of study dummies	Yes	Yes	Yes	Yes
University dummies	Yes	Yes	Yes	Yes
Regional dummies	Yes	Yes	No	No
Region [*] education [*] gender dummies	No	No	Yes	Yes
Ν	24,417	24,417	$24,\!417$	$24,\!417$
R^2	0.042	0.087	0.047	0.087

Table 2. Probability of university drop-out, 2007-2011

Notes. Robust standard errors reported in parenthesis, clustered at the regional level.

Individual controls: gender, high school grade, father with university degree, father with high school degree,

private university, studying in a different region, coming from a disadvantaged family.

Cross-sectional weights used in all specifications. Significance levels: *** = 1%, ** = 5%, * = 10%.

	-	
	(1)	(2)
	Women	Men
Adult male unemployment rate	0.008**	-0.002
	(0.003)	(0.004)
Youth unemployment rate	-0.003*	-0.005**
	(0.002)	(0.002)
Individual controls	Yes	Yes
Cohort dummies	Yes	Yes
Field of study dummies	Yes	Yes
University dummies	Yes	Yes
Regional dummies	Yes	Yes
Ν	14,658	9,759
R^2	0.071	0.103

Table 3. Probability of university drop-out, by gender

Notes. Robust standard errors reported in parenthesis, clustered at the regional level.

Individual controls: high school grade, father with university degree, father with high school degree, private university, studying in a different region, coming from a disadvantaged family.

 $\label{eq:cross-sectional weights used in all specifications. Significance levels: *** = 1\%, ** = 5\%, * = 10\%.$

· · ·	, , , ,	
	(1)	(2)
	Low	High
	school grade	school grade
Adult male unemployment rate	0.008*	0.003
	(0.004)	(0.003)
Youth unemployment rate	-0.003*	-0.001
	(0.002)	(0.0009)
Individual controls	Yes	Yes
Cohort dummies	Yes	Yes
Field of study dummies	Yes	Yes
University dummies	Yes	Yes
Regional dummies	Yes	Yes
Ν	9,426	14,991
\mathbb{R}^2	0.081	0.056

Table 4. Probability of university drop-out, by ability

Notes. Robust standard errors reported in parenthesis, clustered at the regional level.

Individual controls: gender, father with university degree, father with high school degree,

private university, studying in a different region, coming from a disadvantaged family.

Cross-sectional weights used in all specifications. Significance levels: *** = 1%, ** = 5%, * = 10%.

	(1)
Adult male unemployment rate	0.002
	(0.003)
Youth unemployment rate	-0.002*
	(0.001)
Disadvantaged father	-0.003
	(0.011)
Adult male unemployment rate	0.006***
*disadvantaged father	(0.002)
Youth unemployment rate	-0.000
*disadvantaged father	(0.000)
Individual controls	Yes
Cohort dummies	Yes
Field of study dummies	Yes
University dummies	Yes
Regional dummies	Yes
Ν	24,417
\mathbb{R}^2	0.086

Table 5. Probability of university drop-out, by paternal socioeconomic status

Individual controls: gender, high school grade, father with university degree, father with high school degree,

private university, studying in a different region.

Disadvantaged father: absent, unqualified worker (active or retired), or unemployed

Cross-sectional weights used in all specifications. Significance levels: *** = 1%, ** = 5%, * = 10%.

	(1)	(2)
	Science	Humanities
Adult male unemployment rate	-0.006*	0.011**
	(0.003)	(0.005)
Youth unemployment rate	-0.003	-0.003**
	(0.002)	(0.001)
Individual controls	Yes	Yes
Cohort dummies	Yes	Yes
Field of study dummies	Yes	Yes
University dummies	Yes	Yes
Regional dummies	Yes	Yes
Ν	7,641	8,992
R^2	0.121	0.089

Table 6. Probability of university drop-out, by field of study

Individual controls: gender, high school grade, father with university degree, father with high school degree,

private university, studying in a different region, coming from a disadvantaged family.

 $\label{eq:cross-sectional weights used in all specifications. Significance levels: *** = 1\%, ** = 5\%, * = 10\%.$

	(1)	(2)	(3)
	University ranking	University ranking	University ranking
	<25th percentile	25th-75th percentile	>75th percentile
Adult male unemployment rate	0.014**	0.008**	-0.000
	(0.006)	(0.003)	(0.011)
Youth unemployment rate	-0.003	-0.000	0.002
	(0.002)	(0.002)	(0.003)
Individual controls	Yes	Yes	Yes
Cohort dummies	Yes	Yes	Yes
Field of study dummies	Yes	Yes	Yes
University dummies	Yes	Yes	Yes
Regional dummies	Yes	Yes	Yes
Ν	4,567	9,962	$5,\!469$
R^2	0.130	0.089	0.091

Table 7. Probability of university drop-out, by quality of the university

Notes. Robust standard errors reported in parenthesis, clustered at the regional level.

Individual controls: gender, high school grade, father with university degree, father with high school degree,

private university, studying in a different region, coming from a disadvantaged family.

Cross-sectional weights used in all specifications. Significance levels: *** = 1%, ** = 5%, * = 10%.

	(1)	(2)
	Women	Men
Adult male unemployment rate	-0.005	-0.002
	(0.004)	(0.005)
Youth unemployment rate	-0.003	-0.009***
	(0.002)	(0.002)
Adult male unemployment rate	0.015***	0.002
*Humanities	(0.004)	(0.006)
Youth unemployment rate	-0.001	0.000
*Humanities	(0.000)	(0.000)
Individual controls	Yes	Yes
Cohort dummies	Yes	Yes
Field of study dummies	Yes	Yes
University dummies	Yes	Yes
Regional dummies	Yes	Yes
Ν	$9{,}531$	7,102
R^2	0.075	0.116

Table 8a. Probability of university drop-out, by gender and field of study

Individual controls: gender, high school grade, father with university degree, father with high school degree,

private university, studying in a different region, coming from a disadvantaged family.

Cross-sectional weights used in all specifications. Significance levels: *** = 1%, ** = 5%, * = 10%.

	(1)	(2)
	Women	Men
Adult male unemployment rate	0.008**	-0.001
	(0.003)	(0.005)
Youth unemployment rate	-0.003*	-0.005**
	(0.002)	(0.002)
Mother university graduate	-0.019	-0.046***
	(0.017)	(0.022)
Adult male unemployment rate	-0.005*	-0.006
*mother university graduate	(0.003)	(0.009)
Youth unemployment rate	0.000	0.001
*mother university graduate	(0.000)	(0.000)
Individual controls	Yes	Yes
Cohort dummies	Yes	Yes
Field of study dummies	Yes	Yes
University dummies	Yes	Yes
Regional dummies	Yes	Yes
Ν	14,643	9,736
R^2	0.071	0.105

Table 8b. Probability of university drop-out, by gender and mother's education

Individual controls: gender, high school grade, father with university degree, father with high school degree,

private university, studying in a different region, coming from a disadvantaged family.

Cross-sectional weights used in all specifications. Significance levels: *** = 1%, ** = 5%, * = 10%.

	(1)	(2)
Adult male unemployment rate	0.016***	0.001
	(0.002)	(0.002)
Youth unemployment rate	-0.004***	-0.001*
	(0.0004)	(0.001)
Individual controls	No	Yes
Cohort dummies	Yes	Yes
Field of study dummies	Yes	Yes
University dummies	Yes	Yes
Provincial dummies	Yes	Yes
Ν	$23,\!558$	$23,\!549$
\mathbf{R}^2	0.046	0.095

Table 9. Probability of university drop-out, provincial level

Notes. Robust standard errors reported in parenthesis, clustered at the regional level.

Individual controls: gender, high school grade, father with university degree, father with high school degree,

private university, studying in a different region, coming from a disadvantaged family.

Cross-sectional weights used in all specifications. Significance levels: *** = 1%, ** = 5%, * = 10%.

	- / -	
	(1)	(2)
	Placebo	Benchmark
Adult male unemployment rate	-0.000	0.005***
	(0.002)	(0.002)
Youth unemployment rate	0.000	-0.002*
	(0.001)	(0.001)
Individual controls	Yes	Yes
Cohort dummies	Yes	Yes
Field of study dummies	No	No
University dummies	No	No
Regional dummies	Yes	Yes
Ν	20,895	24,961
\mathbb{R}^2	0.073	0.070

Table 10. Probability of university drop-out, placebo

Notes. Robust standard errors reported in parenthesis, clustered at the regional level.

Individual controls: gender, high school grade, father with university degree, father with high school degree,

coming from a disadvantaged family.

 $\label{eq:cross-sectional weights used in all specifications. Significance levels: *** = 1\%, ** = 5\%, * = 10\%.$

<i>v v</i> 1	,		
	(1)	(2)	
Adult male unemployment rate	-0.003	0.001	
	(0.002)	(0.003)	
Youth unemployment rate	0.000	-0.000	
	(0.000)	(0.000)	
Individual controls	Yes	Yes	
Cohort dummies	Yes	Yes	
Field of study dummies	No	No	
University dummies	No	No	
Regional dummies	Yes	No	
Region [*] education [*] gender dummies	No	Yes	
Ν	20,895	20,895	
\mathbb{R}^2	0.073	0.074	

Table 1	1.	Probability	of	university	drop-out.	2004-2007
TODIO I	-	LIODUDIIIUy	OI.	uni voi bioi v	arop out,	2001 2001

Individual controls: gender, high school grade, father with university degree, father with high school degree,

coming from a disadvantaged family.

 $\label{eq:cross-sectional weights used in all specifications. Significance levels: *** = 1\%, ** = 5\%, * = 10\%.$

	(1)
Adult male unemployment rate	-0.002
	(0.002)
Youth unemployment rate	-0.006***
	(0.001)
Adult male unemployment rate	0.010***
*Recession	(0.003)
Youth unemployment rate	-0.003***
*Recession	(0.001)
Recession dummy	Yes
Individual controls	Yes
Field of study dummies	No
University dummies	No
Regional dummies	No
Ν	31,646
\mathbb{R}^2	0.068

Table 12. Probability of university drop-out, 2004-2007-2011

Individual controls: gender, high school grade, father with university degree, father with high school degree,

coming from a disadvantaged family.

Cross-sectional weights used in all specifications. Significance levels: *** = 1%, ** = 5%, * = 10%.

	(1)	(2)
Adult male unemployment rate	0.004*	0.004*
	(0.002)	(0.002)
Youth unemployment rate	-0.002*	-0.002*
	(0.001)	(0.001)
University employability premium	-0.0003	
	(0.0008)	
University wage premium		0.00005
		(0.00005)
Individual controls	Yes	Yes
Cohort dummies	Yes	Yes
Field of study dummies	Yes	Yes
University dummies	Yes	Yes
Regional dummies	Yes	Yes
Ν	$24,\!417$	24,417
R^2	0.087	0.087

Table 13. Probability of university drop-out, university premium

Individual controls: gender, high school grade, father with university degree, father with high school degree,

private university, studying in a different region, coming from a disadvantaged family.

Cross-sectional weights used in all specifications. Significance levels: *** = 1%, ** = 5%, * = 10%.

	(1)	(2)	(3)	(4)	(5)
		Choice of university		University ranking	
	A 11	Low	High	Below	Above
	All	motivation	motivation	median	median
Adult unemployment rate	0.004	-0.002	0.008**	0.003	0.006
	(0.003)	(0.004)	(0.003)	(0.004)	(0.005)
Youth unemployment rate	-0.002	-0.000	-0.002	-0.004*	0.000
	(0.002)	(0.002)	(0.002)	(0.002)	(0.003)
Individual controls	Yes	Yes	Yes	Yes	Yes
Cohort dummies	Yes	Yes	Yes	Yes	Yes
Field of study dummies	Yes	Yes	Yes	Yes	Yes
University dummies	Yes	Yes	Yes	Yes	Yes
Regional dummies	Yes	Yes	Yes	Yes	Yes
Ν	21,236	10,605	10,631	8,420	8,802
R^2	0.168	0.155	0.196	0.160	0.200

Table 14. Other outcomes, probability of on-time graduation

Individual controls: gender, high school grade, father with university degree, father with high school degree,

private university, studying in a different region, coming from a disadvantaged family.

Controlling for drop-outs.

Cross-sectional weights used in all specifications. Significance levels: *** = 1%, ** = 5%, * = 10%.

	(1)
Adult male unemployment rate	0.001
	(0.003)
Youth unemployment rate	0.000
	(0.001)
Individual controls	Yes
Cohort dummies	Yes
Field of study dummies	Yes
University dummies	Yes
Regional dummies	Yes
Ν	$19,\!485$
R^2	0.041

Table 15. Other outcomes, regularity of attendance

Notes. Robust standard errors reported in parenthesis, clustered at the regional level.

Individual controls: gender, high school grade, father with university degree, father with high school degree,

private university, studying in a different region, coming from a disadvantaged family.

Cross-sectional weights used in all specifications. Significance levels: *** = 1%, ** = 5%, * = 10%.

Appendix

	LPM
Adult male unemployment rate	0.004*
	(0.002)
Youth unemployment rate	-0.002**
	(0.001)
Gender	-0.033***
	(0.009)
High school grade	-0.005***
	(0.000)
With father university graduate	-0.081***
	(0.012)
With father high school graduate	-0.042***
	(0.009)
Study in a different region	-0.013**
	(0.005)
From a disadvantaged family	0.033***
	(0.007)
Private university	-0.474*
	(0.026)
Cohort dummies	Yes
Field of study dummies	Yes
University dummies	Yes
Regional dummies	Yes
Ν	24,417
\mathbb{R}^2	0.087

Table A1. Probability of university drop-out, full specification

Notes. Robust standard errors reported in parenthesis clustered at the regional level.

Cross-sectional weights used in all specifications

Significance levels: *** = 1%, ** = 5%, * = 10%.

	- /			
	(1a)	(1b)	(2a)	(2b)
Adult male unemployment rate	0.019***		0.004*	
	(0.002)		(0.002)	
Youth unemployment rate		-0.008***		-0.002*
		(0.001)		(0.001)
Individual controls	No	No	Yes	Yes
Cohort dummies	Yes	Yes	Yes	Yes
Field of study dummies	Yes	Yes	Yes	Yes
University dummies	Yes	Yes	Yes	Yes
Regional dummies	Yes	Yes	Yes	Yes
Ν	24,417	24,417	24,417	24,417
\mathbb{R}^2	0.037	0.031	0.087	0.087

Table A2. Probability of university drop-out, 2007-2011

Notes. Robust standard errors reported in parenthesis, clustered at the regional level

Individual controls: gender, high school grade, father with university degree, father with high school degree,

private university, studying in a different region, coming from a disadvantaged family.

Cross-sectional weights used in all specifications. Significance levels: *** = 1%, ** = 5%, * = 10%.

Survey year year of graduation	$\underset{2004}{2007}$	$\underset{2007}{2011}$
Employability premium		
Youth unemployment rate of high school graduates (3-year average)	20.05	22.41
	(12.09)	(11.17)
Youth unemployment rate of university graduates (3-year average)	21.35	18.22
	(11.88)	(9.91)
Wage premium		
Net monthly wage of high school graduates (in euros)	982	983
	(157)	(170)
Net monthly wage of university graduates (in euros)	$1,\!217$	1,313
	(151)	(163)
Notes: corrected for the survey design using the corresponding weights.		

Table A3. University premium, Mean (standard error)



Figure A1. Change in the adult male unemployment rate by education between 2005-2007

Note: darker shaded areas represent regions that experienced higher increases in the unemployment rate

Figure A2. Change in the youth unemployment rate by gender between 2005-2007 and 2008-2010



Note: darker shaded areas represent regions that experienced higher increases in the unemployment rate

References

Aguiar, M., E. Hurst, and L. Karabarbounis (2013). "Time Use During the Recession," *American Economic Review*, Vol. 103, 1664-1696.

Angrist, J., and S. Pischke (2009). *Mostly Harmless Econometrics: An Empiricist's Companion*, Princeton University Press.

ANVUR (2014). Rapporto sullo Stato del Sistema Universitario e della Ricerca, ANVUR.

Becker, G. (1964). Human Capital: A Theoretical and Empirical Analysis, with Special Reference to Education, New York: Columbia University Press.

Belley, P., and L. Lochner (2007). "The Changing Role of Family Income and Ability in Determining Educational Achievement," *Journal of Human Capital*, Vol. 1, 37–89.

Betts, J., and L. McFarland (1995). "Safe Port in a Storm: The Impact of Labour Market Conditions on Community College enrolments," *Journal of Human Resources*, Vol. 30, 741-765.

Bezemer, D. (2009). "No One Saw This Coming. Understanding Financial Crisis Through Accounting Models," *Research Report 09002*, University of Groningen, Research Institute SOM.

Bordo, M. (2008). "An Historical Perspective on the Crisis of 2007-2008," *NBER* Working Paper No. 14569.

Bound, J., M. Lovenheim, and S. Turner (2010). "Why Have College Completion Rates Declined? An Analysis of Changing Student Preparation and Collegiate Resources," *American Economic Journal: Applied Economics*, Vol. 2, 129-57.

Bratti, M., C. Broccolini, and S. Staffolani (2006). "Is '3+2' Equal to 4? University Reform and Student Academic Performance in Italy," *Working Paper No 251*, Università Politecnica delle Marche I.

Brown, S., and J. Sessions (1999). "Education and Employment Status: A Test of the Strong Screening Hypothesis in Italy," *Economics of Education Review*, Vol. 18, 397-404. Caivano, M., L. Rodano, and S. Siviero (2010). "La Trasmissione della Crisi Finanziaria Globale all'Economia Italiana. Un'Indagine Controfattuale, 2008-2010," *Questioni di Economia e Finanza (Occasional Papers) 64*, Bank of Italy, Economic Research and International Relations Area.

Cameron, S., and C. Taber (2004). "Estimation of Educational Borrowing Constraints Using Returns to Schooling," *Journal of Political Economy*, Vol. 112, 132– 82.

Cardoso, A., E. Fontainha, and C. Monfardini (2010). "Children's and Parents' Time Use: Empirical Evidence on Investment in Human Capital in France, Germany, and Italy," *Review of Economics of the Household*, Vol. 8, 479-504.

Carneiro, P., and J. Heckman (2002). "The Evidence on Credit Constraints in Post–secondary Schooling," *Economic Journal*, Vol. 112, 705-734.

Christian, M. (2007). "Liquidity Constraints and the Cyclicality of College enrolment in the United States," *Oxford Economic Papers*, Vol. 59, 141-169.

Ciccone, A., and E. Papaioannou (2009). "Human Capital, the Structure of Production, and Growth," *Review of Economics and Statistics*, Vol. 91, 66-82.

Cingano, F., and P. Cipollone (2007). "University Drop-out. The Case of Italy," *Temi di discussione (Working papers) 626*, Bank of Italy.

Clark, D. (2011). "Do Recessions Keep Students in School? The Impact of Youth Unemployment on Enrolment in Port-compulsory Education in England," *Economica*, Vol. 78, 523-545.

D'Amuri, F. (2011). "The Impact of the Recession on the Italian Labour Market," *Research in Labour Economics*, Vol. 32, 155–179.

Dellas, H., and P. Sakellaris (2003). "On the Cyclicality of Schooling: Theory and Evidence," *Oxford Economic Papers*, Vol. 55, 148-172.

Di Pietro, G. (2006). "Regional Labour Market Conditions and University Dropout Rates: Evidence from Italy," *Regional Studies*, Vol. 40, 617-630.

Hershbein, B. (2012). "Graduating High School in a Recession: Work, Education, and Home Production," *B.E. Journal of Economic Analysis and Policy*, Vol. 12, 1-30.

Kahn, L. (2010). "The Long-Term Labour Market Consequences of Graduating from College in a Bad Economy," *Labour Economics*, Vol. 17, 303-316.

Long, B. (2014). "The Financial Crisis and Declining College Affordability: How Have Students and Their Families Responded?" In *How the Recession Affected Higher Education*, edited by J. Brown and C. Hoxby, University of Chicago Press.

Lovenheim, M. (2011). "The Effect of Liquid Housing Wealth on College enrolment," *Journal of Labour Economics*, Vol. 29, 741-771.

Lovenheim, M. and L. Reynolds (2012). "The Effect of Housing Wealth on College Choice: Evidence from the Housing Boom," *Journal of Human Resources*, Vol. 48, 1-35.

Mocetti, S. (2012). "Educational Choices and the Selection Process: Before and after Compulsory Schooling," *Education Economics*, Vol. 20, 189-209.

Oreopoulos, P., T. von Wachter, and A. Heisz (2012). "The Short- and Long-Term Career Effects of Graduating in a Recession," *American Economic Journal: Applied Economics*, Vol. 4, 1-29.

Petrongolo, B. and M. San Segundo (2002). "Staying-on at School at 16: the Impact of Labour Market Conditions in Spain," *Economics of Education Review*, Vol 21, 353-365.

Rizzica, L. (2013). "Home or Away? Gender Differences in the Effects of an Expansion of Tertiary Education Supply," *Questioni di Economia e Finanza (Occasional Papers) 181*, Bank of Italy, Economic Research and International Relations Area.

Sestito, P., and M. Tonello (2011). "I Differenziali nella Qualità degli Iscritti alle Università Italiane: il Caso delle Facoltà di Medicina e Chirurgia," *Questioni di Economia e Finanza (Occasional Papers) 90*, Bank of Italy, Economic Research and International Relations Area.

Stinebrickner, R. and T. Stinebrickner (2008). "The Effect of Credit Constraints on the College Drop-Out Decision: A Direct Approach Using a New Panel Study," *American Economic Review*, Vol. 98, 2163-84. Stinebrickner, T. and R. Stinebrickner (2012). "Learning about Academic Ability and the College Dropout Decision," *Journal of Labour Economics*, Vol. 30, 707 -748.

- N. 945 *Simple banking: profitability and the yield curve*, by Piergiorgio Alessandri and Benjamin Nelson (January 2014).
- N. 946 Information acquisition and learning from prices over the business cycle, by Taneli Mäkinen and Björn Ohl (January 2014).
- N. 947 *Time series models with an EGB2 conditional distribution*, by Michele Caivano and Andrew Harvey (January 2014).
- N. 948 *Trade and finance: is there more than just 'trade finance'? Evidence from matched bank-firm data*, by Silvia Del Prete and Stefano Federico (January 2014).
- N. 949 *Natural disasters, growth and institutions: a tale of two earthquakes*, by Guglielmo Barone and Sauro Mocetti (January 2014).
- N. 950 The cost of firms' debt financing and the global financial crisis, by Daniele Pianeselli and Andrea Zaghini (February 2014).
- N. 951 On bank credit risk: systemic or bank-specific? Evidence from the US and UK, by Junye Li and Gabriele Zinna (February 2014).
- N. 952 School cheating and social capital, by Marco Paccagnella and Paolo Sestito (February 2014).
- N. 953 The impact of local minimum wages on employment: evidence from Italy in the 1950s, by Guido de Blasio and Samuele Poy (March 2014).
- N. 954 *Two EGARCH models and one fat tail*, by Michele Caivano and Andrew Harvey (March 2014).
- N. 955 *My parents taught me. Evidence on the family transmission of values*, by Giuseppe Albanese, Guido de Blasio and Paolo Sestito (March 2014).
- N. 956 Political selection in the skilled city, by Antonio Accetturo (March 2014).
- N. 957 *Calibrating the Italian smile with time-varying volatility and heavy-tailed models*, by Michele Leonardo Bianchi (April 2014).
- N. 958 *The intergenerational transmission of reading: is a good example the best sermon?*, by Anna Laura Mancini, Chiara Monfardini and Silvia Pasqua (April 2014).
- N. 959 A tale of an unwanted outcome: transfers and local endowments of trust and cooperation, by Antonio Accetturo, Guido de Blasio and Lorenzo Ricci (April 2014).
- N. 960 *The impact of R&D subsidies on firm innovation*, by Raffaello Bronzini and Paolo Piselli (April 2014).
- N. 961 *Public expenditure distribution, voting, and growth,* by Lorenzo Burlon (April 2014).
- N. 962 *Cooperative R&D networks among firms and public research institutions*, by Marco Marinucci (June 2014).
- N. 963 *Technical progress, retraining cost and early retirement,* by Lorenzo Burlon and Montserrat Vilalta-Bufí (June 2014).
- N. 964 Foreign exchange reserve diversification and the "exorbitant privilege", by Pietro Cova, Patrizio Pagano and Massimiliano Pisani (July 2014).
- N. 965 *Behind and beyond the (headcount) employment rate*, by Andrea Brandolini and Eliana Viviano (July 2014).
- N. 966 Bank bonds: size, systemic relevance and the sovereign, by Andrea Zaghini (July 2014).
- N. 967 Measuring spatial effects in presence of institutional constraints: the case of Italian Local Health Authority expenditure, by Vincenzo Atella, Federico Belotti, Domenico Depalo and Andrea Piano Mortari (July 2014).
- N. 968 Price pressures in the UK index-linked market: an empirical investigation, by Gabriele Zinna (July 2014).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- S. DI ADDARIO, *Job search in thick markets*, Journal of Urban Economics, v. 69, 3, pp. 303-318, **TD No.** 605 (December 2006).
- F. SCHIVARDI and E. VIVIANO, *Entry barriers in retail trade*, Economic Journal, v. 121, 551, pp. 145-170, **TD** No. 616 (February 2007).
- G. FERRERO, A. NOBILI and P. PASSIGLIA, Assessing excess liquidity in the Euro Area: the role of sectoral distribution of money, Applied Economics, v. 43, 23, pp. 3213-3230, **TD No. 627 (April 2007).**
- P. E. MISTRULLI, Assessing financial contagion in the interbank market: maximum entropy versus observed interbank lending patterns, Journal of Banking & Finance, v. 35, 5, pp. 1114-1127, TD No. 641 (September 2007).
- E. CIAPANNA, *Directed matching with endogenous markov probability: clients or competitors?*, The RAND Journal of Economics, v. 42, 1, pp. 92-120, **TD No. 665 (April 2008).**
- M. BUGAMELLI and F. PATERNÒ, *Output growth volatility and remittances*, Economica, v. 78, 311, pp. 480-500, **TD No. 673 (June 2008).**
- V. DI GIACINTO e M. PAGNINI, Local and global agglomeration patterns: two econometrics-based indicators, Regional Science and Urban Economics, v. 41, 3, pp. 266-280, **TD No. 674 (June 2008)**.
- G. BARONE and F. CINGANO, Service regulation and growth: evidence from OECD countries, Economic Journal, v. 121, 555, pp. 931-957, TD No. 675 (June 2008).
- P. SESTITO and E. VIVIANO, *Reservation wages: explaining some puzzling regional patterns*, Labour, v. 25, 1, pp. 63-88, **TD No. 696 (December 2008).**
- R. GIORDANO and P. TOMMASINO, *What determines debt intolerance? The role of political and monetary institutions*, European Journal of Political Economy, v. 27, 3, pp. 471-484, **TD No. 700 (January 2009).**
- P. ANGELINI, A. NOBILI and C. PICILLO, *The interbank market after August 2007: What has changed, and why?*, Journal of Money, Credit and Banking, v. 43, 5, pp. 923-958, **TD No. 731 (October 2009).**
- G. BARONE and S. MOCETTI, *Tax morale and public spending inefficiency*, International Tax and Public Finance, v. 18, 6, pp. 724-49, **TD No. 732 (November 2009).**
- L. FORNI, A. GERALI and M. PISANI, *The Macroeconomics of Fiscal Consolidation in a Monetary Union:* the Case of Italy, in Luigi Paganetto (ed.), Recovery after the crisis. Perspectives and policies, VDM Verlag Dr. Muller, **TD No. 747 (March 2010).**
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap changes before and during the subprime financial turmoil, in Barbara L. Campos and Janet P. Wilkins (eds.), The Financial Crisis: Issues in Business, Finance and Global Economics, New York, Nova Science Publishers, Inc., TD No. 749 (March 2010).
- A. LEVY and A. ZAGHINI, *The pricing of government guaranteed bank bonds*, Banks and Bank Systems, v. 6, 3, pp. 16-24, **TD No. 753 (March 2010).**
- G. BARONE, R. FELICI and M. PAGNINI, *Switching costs in local credit markets,* International Journal of Industrial Organization, v. 29, 6, pp. 694-704, **TD No. 760 (June 2010).**
- G. BARBIERI, C. ROSSETTI e P. SESTITO, The determinants of teacher mobility: evidence using Italian teachers' transfer applications, Economics of Education Review, v. 30, 6, pp. 1430-1444, TD No. 761 (marzo 2010).
- G. GRANDE and I. VISCO, *A public guarantee of a minimum return to defined contribution pension scheme members*, The Journal of Risk, v. 13, 3, pp. 3-43, **TD No. 762 (June 2010).**
- P. DEL GIOVANE, G. ERAMO and A. NOBILI, Disentangling demand and supply in credit developments: a survey-based analysis for Italy, Journal of Banking and Finance, v. 35, 10, pp. 2719-2732, TD No. 764 (June 2010).
- G. BARONE and S. MOCETTI, With a little help from abroad: the effect of low-skilled immigration on the female labour supply, Labour Economics, v. 18, 5, pp. 664-675, **TD No. 766 (July 2010).**
- S. FEDERICO and A. FELETTIGH, *Measuring the price elasticity of import demand in the destination markets of italian exports,* Economia e Politica Industriale, v. 38, 1, pp. 127-162, **TD No. 776 (October 2010).**
- S. MAGRI and R. PICO, *The rise of risk-based pricing of mortgage interest rates in Italy*, Journal of Banking and Finance, v. 35, 5, pp. 1277-1290, **TD No. 778 (October 2010).**

- M. TABOGA, Under/over-valuation of the stock market and cyclically adjusted earnings, International Finance, v. 14, 1, pp. 135-164, **TD No. 780 (December 2010).**
- S. NERI, *Housing, consumption and monetary policy: how different are the U.S. and the Euro area?*, Journal of Banking and Finance, v.35, 11, pp. 3019-3041, **TD No. 807 (April 2011).**
- V. CUCINIELLO, *The welfare effect of foreign monetary conservatism with non-atomistic wage setters*, Journal of Money, Credit and Banking, v. 43, 8, pp. 1719-1734, **TD No. 810 (June 2011).**
- A. CALZA and A. ZAGHINI, welfare costs of inflation and the circulation of US currency abroad, The B.E. Journal of Macroeconomics, v. 11, 1, Art. 12, **TD No. 812 (June 2011).**
- I. FAIELLA, *La spesa energetica delle famiglie italiane*, Energia, v. 32, 4, pp. 40-46, **TD No. 822 (September 2011).**
- D. DEPALO and R. GIORDANO, *The public-private pay gap: a robust quantile approach*, Giornale degli Economisti e Annali di Economia, v. 70, 1, pp. 25-64, **TD No. 824 (September 2011).**
- R. DE BONIS and A. SILVESTRINI, *The effects of financial and real wealth on consumption: new evidence from* OECD countries, Applied Financial Economics, v. 21, 5, pp. 409–425, **TD No. 837 (November 2011).**
- F. CAPRIOLI, P. RIZZA and P. TOMMASINO, *Optimal fiscal policy when agents fear government default*, Revue Economique, v. 62, 6, pp. 1031-1043, **TD No. 859 (March 2012).**

2012

- F. CINGANO and A. ROSOLIA, *People I know: job search and social networks*, Journal of Labor Economics, v. 30, 2, pp. 291-332, **TD No. 600 (September 2006).**
- G. GOBBI and R. ZIZZA, Does the underground economy hold back financial deepening? Evidence from the italian credit market, Economia Marche, Review of Regional Studies, v. 31, 1, pp. 1-29, TD No. 646 (November 2006).
- S. MOCETTI, *Educational choices and the selection process before and after compulsory school*, Education Economics, v. 20, 2, pp. 189-209, **TD No. 691 (September 2008).**
- P. PINOTTI, M. BIANCHI and P. BUONANNO, *Do immigrants cause crime?*, Journal of the European Economic Association, v. 10, 6, pp. 1318–1347, **TD No. 698 (December 2008).**
- M. PERICOLI and M. TABOGA, *Bond risk premia, macroeconomic fundamentals and the exchange rate*, International Review of Economics and Finance, v. 22, 1, pp. 42-65, **TD No. 699 (January 2009).**
- F. LIPPI and A. NOBILI, *Oil and the macroeconomy: a quantitative structural analysis,* Journal of European Economic Association, v. 10, 5, pp. 1059-1083, **TD No. 704 (March 2009).**
- G. ASCARI and T. ROPELE, *Disinflation in a DSGE perspective: sacrifice ratio or welfare gain ratio?*, Journal of Economic Dynamics and Control, v. 36, 2, pp. 169-182, **TD No. 736 (January 2010).**
- S. FEDERICO, *Headquarter intensity and the choice between outsourcing versus integration at home or abroad*, Industrial and Corporate Chang, v. 21, 6, pp. 1337-1358, **TD No. 742 (February 2010).**
- I. BUONO and G. LALANNE, *The effect of the Uruguay Round on the intensive and extensive margins of trade*, Journal of International Economics, v. 86, 2, pp. 269-283, **TD No. 743 (February 2010).**
- A. BRANDOLINI, S. MAGRI and T. M SMEEDING, Asset-based measurement of poverty, In D. J. Besharov and K. A. Couch (eds), Counting the Poor: New Thinking About European Poverty Measures and Lessons for the United States, Oxford and New York: Oxford University Press, TD No. 755 (March 2010).
- S. GOMES, P. JACQUINOT and M. PISANI, The EAGLE. A model for policy analysis of macroeconomic interdependence in the euro area, Economic Modelling, v. 29, 5, pp. 1686-1714, TD No. 770 (July 2010).
- A. ACCETTURO and G. DE BLASIO, Policies for local development: an evaluation of Italy's "Patti Territoriali", Regional Science and Urban Economics, v. 42, 1-2, pp. 15-26, TD No. 789 (January 2006).
- F. BUSETTI and S. DI SANZO, *Bootstrap LR tests of stationarity, common trends and cointegration,* Journal of Statistical Computation and Simulation, v. 82, 9, pp. 1343-1355, **TD No. 799 (March 2006).**
- S. NERI and T. ROPELE, *Imperfect information, real-time data and monetary policy in the Euro area,* The Economic Journal, v. 122, 561, pp. 651-674, **TD No. 802 (March 2011).**
- A. ANZUINI and F. FORNARI, *Macroeconomic determinants of carry trade activity*, Review of International Economics, v. 20, 3, pp. 468-488, **TD No. 817 (September 2011).**
- M. AFFINITO, *Do interbank customer relationships exist? And how did they function in the crisis? Learning from Italy*, Journal of Banking and Finance, v. 36, 12, pp. 3163-3184, **TD No. 826 (October 2011).**

- P. GUERRIERI and F. VERGARA CAFFARELLI, Trade Openness and International Fragmentation of Production in the European Union: The New Divide?, Review of International Economics, v. 20, 3, pp. 535-551, TD No. 855 (February 2012).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Network effects of public transposrt infrastructure: evidence on Italian regions, Papers in Regional Science, v. 91, 3, pp. 515-541, TD No. 869 (July 2012).
- A. FILIPPIN and M. PACCAGNELLA, *Family background, self-confidence and economic outcomes,* Economics of Education Review, v. 31, 5, pp. 824-834, **TD No. 875 (July 2012).**

2013

- A. MERCATANTI, A likelihood-based analysis for relaxing the exclusion restriction in randomized experiments with imperfect compliance, Australian and New Zealand Journal of Statistics, v. 55, 2, pp. 129-153, TD No. 683 (August 2008).
- F. CINGANO and P. PINOTTI, *Politicians at work. The private returns and social costs of political connections*, Journal of the European Economic Association, v. 11, 2, pp. 433-465, **TD No. 709 (May 2009).**
- F. BUSETTI and J. MARCUCCI, *Comparing forecast accuracy: a Monte Carlo investigation*, International Journal of Forecasting, v. 29, 1, pp. 13-27, **TD No. 723 (September 2009).**
- D. DOTTORI, S. I-LING and F. ESTEVAN, *Reshaping the schooling system: The role of immigration*, Journal of Economic Theory, v. 148, 5, pp. 2124-2149, **TD No. 726 (October 2009).**
- A. FINICELLI, P. PAGANO and M. SBRACIA, *Ricardian Selection*, Journal of International Economics, v. 89, 1, pp. 96-109, **TD No. 728 (October 2009).**
- L. MONTEFORTE and G. MORETTI, *Real-time forecasts of inflation: the role of financial variables*, Journal of Forecasting, v. 32, 1, pp. 51-61, **TD No. 767 (July 2010).**
- R. GIORDANO and P. TOMMASINO, *Public-sector efficiency and political culture*, FinanzArchiv, v. 69, 3, pp. 289-316, **TD No. 786 (January 2011).**
- E. GAIOTTI, Credit availablility and investment: lessons from the "Great Recession", European Economic Review, v. 59, pp. 212-227, TD No. 793 (February 2011).
- F. NUCCI and M. RIGGI, *Performance pay and changes in U.S. labor market dynamics*, Journal of Economic Dynamics and Control, v. 37, 12, pp. 2796-2813, **TD No. 800 (March 2011).**
- G. CAPPELLETTI, G. GUAZZAROTTI and P. TOMMASINO, *What determines annuity demand at retirement?*, The Geneva Papers on Risk and Insurance – Issues and Practice, pp. 1-26, **TD No. 805 (April 2011).**
- A. ACCETTURO e L. INFANTE, Skills or Culture? An analysis of the decision to work by immigrant women in Italy, IZA Journal of Migration, v. 2, 2, pp. 1-21, TD No. 815 (July 2011).
- A. DE SOCIO, *Squeezing liquidity in a "lemons market" or asking liquidity "on tap"*, Journal of Banking and Finance, v. 27, 5, pp. 1340-1358, **TD No. 819 (September 2011).**
- S. GOMES, P. JACQUINOT, M. MOHR and M. PISANI, Structural reforms and macroeconomic performance in the euro area countries: a model-based assessment, International Finance, v. 16, 1, pp. 23-44, TD No. 830 (October 2011).
- G. BARONE and G. DE BLASIO, *Electoral rules and voter turnout*, International Review of Law and Economics, v. 36, 1, pp. 25-35, **TD No. 833 (November 2011).**
- O. BLANCHARD and M. RIGGI, Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices, Journal of the European Economic Association, v. 11, 5, pp. 1032-1052, **TD No. 835 (November 2011).**
- R. CRISTADORO and D. MARCONI, *Household savings in China*, in G. Gomel, D. Marconi, I. Musu, B. Quintieri (eds), The Chinese Economy: Recent Trends and Policy Issues, Springer-Verlag, Berlin, TD No. 838 (November 2011).
- E. GENNARI and G. MESSINA, How sticky are local expenditures in Italy? Assessing the relevance of the flypaper effect through municipal data, International Tax and Public Finance (DOI: 10.1007/s10797-013-9269-9), TD No. 844 (January 2012).
- A. ANZUINI, M. J. LOMBARDI and P. PAGANO, *The impact of monetary policy shocks on commodity prices*, International Journal of Central Banking, v. 9, 3, pp. 119-144, **TD No. 851 (February 2012).**
- R. GAMBACORTA and M. IANNARIO, *Measuring job satisfaction with CUB models*, Labour, v. 27, 2, pp. 198-224, **TD No. 852 (February 2012).**

- G. ASCARI and T. ROPELE, Disinflation effects in a medium-scale new keynesian model: money supply rule versus interest rate rule, European Economic Review, v. 61, pp. 77-100, TD No. 867 (April 2012).
- E. BERETTA and S. DEL PRETE, Banking consolidation and bank-firm credit relationships: the role of geographical features and relationship characteristics, Review of Economics and Institutions, v. 4, 3, pp. 1-46, TD No. 901 (February 2013).
- M. ANDINI, G. DE BLASIO, G. DURANTON and W. STRANGE, Marshallian labor market pooling: evidence from Italy, Regional Science and Urban Economics, v. 43, 6, pp.1008-1022, TD No. 922 (July 2013).
- G. SBRANA and A. SILVESTRINI, Forecasting aggregate demand: analytical comparison of top-down and bottom-up approaches in a multivariate exponential smoothing framework, International Journal of Production Economics, v. 146, 1, pp. 185-98, TD No. 929 (September 2013).
- A. FILIPPIN, C. V, FIORIO and E. VIVIANO, *The effect of tax enforcement on tax morale*, European Journal of Political Economy, v. 32, pp. 320-331, **TD No. 937 (October 2013).**

2014

- M. TABOGA, *The riskiness of corporate bonds*, Journal of Money, Credit and Banking, v.46, 4, pp. 693-713, **TD No. 730 (October 2009).**
- G. MICUCCI and P. ROSSI, *Il ruolo delle tecnologie di prestito nella ristrutturazione dei debiti delle imprese in crisi*, in A. Zazzaro (a cura di), Le banche e il credito alle imprese durante la crisi, Bologna, Il Mulino, TD No. 763 (June 2010).
- P. ANGELINI, S. NERI and F. PANETTA, *The interaction between capital requirements and monetary policy*, Journal of Money, Credit and Banking, v. 46, 6, pp. 1073-1112, **TD No. 801 (March 2011).**
- M. FRANCESE and R. MARZIA, Is there Room for containing healthcare costs? An analysis of regional spending differentials in Italy, The European Journal of Health Economics, v. 15, 2, pp. 117-132, TD No. 828 (October 2011).
- L. GAMBACORTA and P. E. MISTRULLI, Bank heterogeneity and interest rate setting: what lessons have we learned since Lehman Brothers?, Journal of Money, Credit and Banking, v. 46, 4, pp. 753-778, TD No. 829 (October 2011).
- M. PERICOLI, *Real term structure and inflation compensation in the euro area*, International Journal of Central Banking, v. 10, 1, pp. 1-42, **TD No. 841 (January 2012).**
- V. DI GACINTO, M. GOMELLINI, G. MICUCCI and M. PAGNINI, *Mapping local productivity advantages in Italy: industrial districts, cities or both?*, Journal of Economic Geography, v. 14, pp. 365–394, TD No. 850 (January 2012).
- A. ACCETTURO, F. MANARESI, S. MOCETTI and E. OLIVIERI, Don't Stand so close to me: the urban impact of immigration, Regional Science and Urban Economics, v. 45, pp. 45-56, TD No. 866 (April 2012).
- S. FEDERICO, *Industry dynamics and competition from low-wage countries: evidence on Italy*, Oxford Bulletin of Economics and Statistics, v. 76, 3, pp. 389-410, **TD No. 879 (September 2012).**
- F. D'AMURI and G. PERI, *Immigration, jobs and employment protection: evidence from Europe before and during the Great Recession,* Journal of the European Economic Association, v. 12, 2, pp. 432-464, TD No. 886 (October 2012).
- M. TABOGA, *What is a prime bank? A euribor-OIS spread perspective*, International Finance, v. 17, 1, pp. 51-75, **TD No. 895 (January 2013).**
- L. GAMBACORTA and F. M. SIGNORETTI, Should monetary policy lean against the wind? An analysis based on a DSGE model with banking, Journal of Economic Dynamics and Control, v. 43, pp. 146-74, **TD No. 921 (July 2013).**
- U. ALBERTAZZI and M. BOTTERO, *Foreign bank lending: evidence from the global financial crisis,* Journal of International Economics, v. 92, 1, pp. 22-35, **TD No. 926 (July 2013).**
- R. DE BONIS and A. SILVESTRINI, *The Italian financial cycle: 1861-2011*, Cliometrica, v.8, 3, pp. 301-334, **TD No. 936 (October 2013).**
- D. PIANESELLI and A. ZAGHINI, *The cost of firms' debt financing and the global financial crisis*, Finance Research Letters, v. 11, 2, pp. 74-83, **TD No. 950 (February 2014).**
- A. ZAGHINI, Bank bonds: size, systemic relevance and the sovereign, International Finance, v. 17, 2, pp. 161-183, TD No. 966 (July 2014).

FORTHCOMING

- M. BUGAMELLI, S. FABIANI and E. SETTE, *The age of the dragon: the effect of imports from China on firmlevel prices*, Journal of Money, Credit and Banking, **TD No. 737 (January 2010).**
- F. D'AMURI, *Gli effetti della legge 133/2008 sulle assenze per malattia nel settore pubblico*, Rivista di Politica Economica, **TD No. 787 (January 2011).**
- E. COCOZZA and P. PISELLI, Testing for east-west contagion in the European banking sector during the financial crisis, in R. Matoušek; D. Stavárek (eds.), Financial Integration in the European Union, Taylor & Francis, TD No. 790 (February 2011).
- R. BRONZINI and E. IACHINI, Are incentives for R&D effective? Evidence from a regression discontinuity approach, American Economic Journal : Economic Policy, TD No. 791 (February 2011).
- G. DE BLASIO, D. FANTINO and G. PELLEGRINI, *Evaluating the impact of innovation incentives: evidence from an unexpected shortage of funds*, Industrial and Corporate Change, **TD No. 792 (February 2011).**
- A. DI CESARE, A. P. STORK and C. DE VRIES, *Risk measures for autocorrelated hedge fund returns*, Journal of Financial Econometrics, **TD No. 831 (October 2011).**
- D. FANTINO, A. MORI and D. SCALISE, Collaboration between firms and universities in Italy: the role of a firm's proximity to top-rated departments, Rivista Italiana degli economisti, TD No. 884 (October 2012).
- G. BARONE and S. MOCETTI, *Natural disasters, growth and institutions: a tale of two earthquakes,* Journal of Urban Economics, **TD No. 949 (January 2014).**